Author's response to reviews

Title: The effects of an area-based intervention on the uptake of maternal and child health assessments in Australia

Authors:

M Kelaher (mkelaher@unimelb.edu.au)
D Dunt (d.dunt@unimelb.edu.au)
P Feldman (feldmanp@unimelb.edu.au)
A Nolan (andrea.nolan@deakin.edu.au)
B Raban (b.raban@unimleb.edu.au)

Version: 2 Date: 26 May 2008

Author's response to reviews:

Thank-you for your useful advice. We have revised the abstract to improve its readability. We are happy to accept the editors advice about where the paper sites best.